



A Survey of the Diagnosis of Peripheral Neuropathy Using Intelligent and Wearable Systems

Muhammad Talha^{1,*}, Maria Kyrarini^{2,*} and Ehsan Ali Buriro¹

- ¹ Department of Electronic Engineering, Quaid-e-Awam University of Science and Technology, Nawabshah 67450, Pakistan; ehsan.ali@quest.edu.pk
- ² Department of Electrical and Computer Engineering, Santa Clara University, Santa Clara, CA 95053, USA
- * Correspondence: talha.09es79@gmail.com (M.T.); mkyrarini@scu.edu (M.K.); Tel.: +92-333-349-2479 (M.T.)

Abstract: In recent years, the usage of wearable systems in healthcare has gained much attention, as they can be easily worn by the subject and provide a continuous source of data required for the tracking and diagnosis of multiple kinds of abnormalities or diseases in the human body. Wearable systems can be made useful in improving a patient's quality of life and at the same time reducing the overall cost of caring for individuals including the elderly. In this survey paper, the recent research in the development of intelligent wearable systems for the diagnosis of peripheral neuropathy is discussed. The paper provides detailed information about recent techniques based on different wearable sensors for the diagnosis of peripheral neuropathy including experimental protocols, biomarkers, and other specifications and parameters such as the type of signals and data processing methods, locations of sensors, the scales and tests used in the study, and the scope of the study. It also highlights challenges that are still present in order to make wearable devices more effective in the diagnosis of peripheral neuropathy in clinical settings.

Keywords: wearable systems; peripheral neuropathy; intelligent systems; sensors



Citation: Talha, M.; Kyrarini, M.; Buriro, E.A. A Survey of the Diagnosis of Peripheral Neuropathy Using Intelligent and Wearable Systems. *Technologies* **2023**, *11*, 163. https://doi.org/10.3390/ technologies11060163

Academic Editor: Mario Munoz-Organero

Received: 18 September 2023 Revised: 31 October 2023 Accepted: 6 November 2023 Published: 17 November 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

One of the major causes of disability worldwide is peripheral neuropathy (PN). It is described as the damage of peripheral nerves that includes both sensory and motor nerves, which can cause severe motor impairments, and 2.4% of the total world population suffers due to the prevalence of peripheral neuropathy [1]. There are various causes of this disease; however, the most affected ones are those people who have a prolonged history of diabetes. Almost half of diabetic patients develop neuropathy at some stage of diabetes [2]. Other causes of peripheral neuropathy include chemotherapy-induced peripheral neuropathy (CIPN) [3], nerve compression, injury, alcohol use, hereditary diseases, toxin exposure, and vitamin deficiencies [4]. Globally, USD 375 billion were spent in 2010 for the treatment of peripheral neuropathy in diabetic patients, and it is expected to reach USD 490 billion by 2030 [5]. The impact of peripheral neuropathy on healthcare costs and quality of life due to pain, gait instability, foot ulceration, amputation, and injury due to risk of fall demand an effective strategy or method for the diagnosis and treatment of the disease as early as possible [6]. Peripheral neuropathy has become the most common neurologic condition that is faced by physicians in almost all medical specialties. The main challenge for physicians is to effectively screen an asymptomatic patient for peripheral neuropathy since the symptoms of peripheral neuropathy are not prominent in its early stages, which makes it difficult to diagnose by physicians and requires careful and expert analysis for the diagnosis process [7,8]. Peripheral neuropathy is the main reason for foot ulceration and amputation in diabetes patients [9]. For this purpose, nerve conduction studies (NCS) along with electromyography (EMG) are the most widely and effectively used tool for the screening of peripheral neuropathy; however, these techniques are time-consuming, costly, and labor intensive, and it is also not practical to use NCS/EMG in clinical settings [10].

In recent years, rapid advancement in the field of information and communication techniques has enabled us to model, design, and produce compact mobile devices with efficient processing capabilities and battery life [11]. The physical sizes of wearable devices have significantly reduced due to modern fabrication techniques, such as microelectromechanical (MEM) techniques, which play a very important role in the improved productivity, and the high accuracy and sensitivity of wearable devices [12]. The concept of wearable technology was first introduced by a mathematics professor "Edward O Thorp" in the early 1960s to predict roulette wheels [13]. Wearable devices are designed with specific electronic functions, and they come in various materials depending upon their use [14]. Due to their feasibility and low cost, applications of wearable devices are growing rapidly in many research areas such as healthcare [15], education [16], sports medicine [17], military purposes [18,19], and social networking [20]. In Ref. [21], different applications of wearable sensors in consumer sports are discussed. Wearable devices are also commercially available for healthcare purposes, such as different commercially available devices from different brands for measuring step counts and heart rates that are compared in [22] with respect to their accuracy. Wearable devices have proven to be reliable in some applications such as step counts and distance measuring. However, further investigation, research, and improved algorithms are needed to make wearable devices more accurate for clinical diagnosis and to explore more applications in the healthcare field [23,24].

This survey paper summarizes the state-of-the-art research work that has been carried out for the diagnosis of peripheral neuropathy using wearable devices that may assist physicians and help maintain the health of patients who are suffering from peripheral neuropathy. This research article has the following aims: (i) presenting recent research techniques based on wearable devices for the diagnosis of peripheral neuropathy, and (ii) highlighting the research gaps and open challenges in the diagnosis of peripheral neuropathy. The main purpose of this survey paper is to present the effectiveness of wearable technology in the diagnosis of peripheral neuropathy, to address the issues that are still present in using wearable devices for the diagnosis of peripheral neuropathy, and to highlight the gap that needs to be addressed in the future in order to make wearable devices more suitable for the diagnosis of peripheral neuropathy. The following questions are answered in this paper:

- 1. What research has been carried out towards developing systems based on wearable devices for the diagnosis of peripheral neuropathy?
- 2. Which types of wearable devices are the most suitable or commonly used for the diagnosis of peripheral neuropathy?
- 3. How can wearable technology assist physicians and contribute to improving the health of patients having peripheral neuropathy or at risk of developing peripheral neuropathy?
- 4. What are the challenges that wearable devices are facing in the diagnosis of peripheral neuropathy?

These questions led us to identify the role of wearable devices in the diagnosis of peripheral neuropathy and to highlight the potential future work that is needed to make wearable devices more accurate and commercially available for the diagnosis of peripheral neuropathy in clinical settings. This paper is organized as follows: Section 2 discusses the criteria for the selection of research papers; Section 3 describes the recent research work in the field of wearable devices for the diagnosis of peripheral neuropathy using only the same type of wearable sensors; Section 4 describes the systems based on multiple types of sensors in order to diagnose PN; Section 5 discusses the research gaps in the field of wearable technology for the diagnosis of peripheral neuropathy; and Section 6 concludes the paper.

2. Methodology

The selection of papers was carried out in three steps: (i) initial search in different digital libraries to identify relevant research papers based on their title and abstract; (ii) filtering out of papers based on defined criteria; and (iii) selection of most appropriate research papers. First, an extensive search was carried out to find research papers that were based on noninvasive techniques for the diagnosis of peripheral neuropathy using wearable technologies on IEEE Xplorer, Science Direct, the ACM digital library, and Google Scholar. These databases were selected, as they possess a collection of indexed publications, journals, and conference proceedings that include wearable technology in healthcare. Research papers that were included from each database were published between the years 2013 and 2023. The following terms were used to search research articles in these databases: peripheral neuropathy and wearable systems/devices/sensors combined in different ways with the words non-invasive, screening, diagnosis, intelligent, automated, and machine learning. The initial search list after removing duplicates included 33 papers from IEEE Xplorer, 12 papers from Science Direct, 10 papers from the ACM digital library, and 21 papers from Google Scholar.

In the next step, the authors reviewed the potentially relevant research papers based on the following criteria: (1) the technique used in the study should be completely noninvasive; (2) only wearable devices/sensors are used for data collection; (3) data acquisition method is provided; (4) physical prototype of the model exists; (5) studies based on wearable systems for the treatment or prediction of risk of falling in already diagnosed peripheral neuropathy patients are not included; and (6) only the literature published in English is included. Finally, 15 research articles were chosen which were purely focused on the diagnosis of peripheral neuropathy using wearable sensors. Among these, four articles focus on gait kinematics using inertial sensors for the diagnosis of PN; three research papers are based on the diagnosis of peripheral neuropathy through the analysis of foot plantar pressure; three studies are related to the diagnosis of peripheral neuropathy using electrocardiography (ECG) data; and five research studies are based on the use of multiple wearable sensors for the detection of peripheral neuropathy. In this paper, systems proposed by researchers for the analysis of PN are divided into two broad categories, which include single-type wearable sensor-based platforms and multiple-type wearable sensor-based platforms. Research platforms based on single-type sensors are subdivided into the following three categories: the first category includes the diagnosis of PN by studying human gait using inertial sensors; the second category includes foot plantar pressure-based systems; and the third category focuses on the analysis of the ECG signal in order to diagnose the damage of nerves responsible for proper heart functioning.

3. Intelligent Wearable Systems Using Single-Sensor Type for Diagnosis of Peripheral Neuropathy

The automated diagnosis of peripheral neuropathy (PN) has been performed in many studies [8,25–34] where only a single sensor or technique is used for the data acquisition purpose. These techniques are based on analyzing foot plantar pressure or gait analysis, vibration- and sensitivity-based confocal microscopic image processing, thermal image processing, ultrasound scanner, and nerve conduction studies (NCS) or electromyography (EMG). However, in order to diagnose peripheral neuropathy, a continuous source of data is necessary to diagnose and track the progression of PN. Image-based approaches can provide a feasible way of detecting PN. In a number of research studies [35–38], a subject's gait is analyzed using cameras, but, due to the camera's limited field of view, it cannot provide continuous data which can result in missed data points [39]. Similarly, other nonwearable techniques such as vibration- or perception-based methods and thermal images provide suitable ways for the detection of PN, but these types of methods are not suitable for long-term data collection which is vital for the diagnosis and progression of PN [40]. For this purpose, wearable sensors are the most suitable way to collect continuous data, as wearable devices are designed in such a way that they can be easily worn by patients and attached to the human body directly [41]. They also provide a way to collect and store patient activity data for later use [42].

In this paper, the research techniques for the diagnosis of peripheral neuropathy using a single-type wearable sensor are divided into three categories depending on the type of sensor, the placement of the sensor, and obtained biomarkers or feature points.

3.1. Wearable Inertial Sensor-Based Intelligent Systems for the Diagnosis of PN

Wearable inertial sensors are mostly used for activity recognition tasks such as human gait or postural positions with potential applications in healthcare and well-being, as they offer reliable and accurate methods for studying human motion [43]. There are different configurations of inertial sensors that can be used for recording and analyzing human gait for the purpose of diagnosis. In most cases, inertial sensors are attached to the leg, foot, or waist of the human body [44]. The most commonly used sensor in most studies is an accelerometer in order to study human motion [45]. The combination of an accelerometer and gyroscope or inertial measurement unit has also recently been used by many researchers [46–49] for studying human motion for different applications.

Chen and Shanshan [50] use a wearable inertial sensor, i.e., accelerometer for the early screening of peripheral neuropathy in diabetic patients and evaluate how much valuable information can be added by the wearable inertial sensor in the screening process of peripheral neuropathy. In this study, the authors focus on developing a wearable system based on inertial sensors that distinguish the gait of diabetic patients with and without peripheral neuropathy. The diagnosis of PN is based on the degradation in gait which results in a slower gait, limited knee and ankle mobility, and shorter steps. The experimental study was performed on 106 participants (aged 38 to 83 years old—54 female and 52 male); among those, 30 were diagnosed as confirmed diabetic peripheral neuropathic patients, and 76 healthy individuals were included. The study was approved by the Ethics Committee of Shanghai Ninth People's Hospital, China. The ground truth in this study was taken from the results of NCS studies and physicians' diagnoses. For data collection, an ear-worn 3three-axis accelerometer was used to capture movements in lateral, forward, and vertical directions at a sampling rate of 100 Hz. During walking trials, each participant had to walk 10 m for three rounds while capturing motion data.

The data were transmitted wirelessly to a tablet in real time. To obtain the gait feature, vertical acceleration was used to segment gait cycles. To identify heel-strike events, a customized peak detection algorithm was used by delineating each gait cycle from a heel-strike event to the instant before the next heel-strike event. The gait features that were used in this research include heel-strike events, toe-off events, variance, skewness, and kurtosis of the gait signal. Other features like the maximum amplitude of the gait signal, cadence per minute, and gait speed during the 10-m walk test were also included for analysis.

After data processing, five categories of logistic regression models (Model A to Model E) were used to predict the presence of peripheral neuropathy using different databases having different feature points. The first model, i.e., Model A, only used the total score from the Michigan neuropathy screening instrument (MNSI) history; the second model was based on physical examination of motor and sensory functions using nerve conduction studies (NCS). Model C used both the data from MNSI history and NCS studies. The fourth model, Model D, only used data from an ear-worn accelerometer, and the last model, Model E, was based on the combination of MNSI history and gait features from the inertial sensor. Age and gender were also used as feature points in all five proposed models. The main reason for creating different models was to quantitatively measure the contribution of wearable sensors to the early screening of PN. In order to compare the performance of the proposed models, the authors applied the likelihood ratio (LR) test to find the best possible gait features for the diagnosis of PN. Significant increased LR test stats show that the new feature or biomarker enhances the performance of the models and improves model fit statistically. To examine overfitting, the Brier score was calculated by observing the difference between the true probability and observed probability. The results of this study show that three gait features can play a great role in distinguishing abnormal walking in PN patients. These features include the skewness of lateral acceleration, the

maximum amplitude of lateral acceleration, and the range normalized maximum amplitude of lateral acceleration. The overall results show that these three gait features can be used to diagnose PN, as they reveal that gait patterns in DPN patients are less extreme and exhibit less sway in the lateral direction. Additionally, step length and gait speed are relatively higher in PN patients compared to healthy individuals. Among all five proposed models in this research, Model E, which consists of inertial data as well as MNSI history, outperformed all other proposed models and suggests that valuable information can be added by wearable sensors in the screening of PN in order to make the diagnosis process more accurate and diagnose the disease in time.

Cohen et al. [51] proposed a framework for the diagnosis of peripheral neuropathy based on an inertial measurement unit (IMU) and tandem walking test. The main focus of this study was to determine whether the tandem walking test can be performed successfully for the screening of peripheral neuropathy. In the experimental study, both healthy subjects and subjects with peripheral neuropathy were recruited. All subjects could walk without any gait aid. The experiment included 21 subjects with PN (13 males, 8 females-age 60 ± 12.4 years) (mean \pm SD) including small fiber, large fiber, or mixed large and small fiber neuropathy, and 61 healthy subjects (31 males, 30 females—age 49.6 \pm 16.0 years). This study was approved by the Institutional Review Board for Human Subjects Research of Baylor College of Medicine, Houston, TX, USA. The ground truth was provided by a medical expert using electromyography (EMG). At the time of the walking trials, subjects wore only socks for hygiene purposes. The walking task consists of a 10-step walk on the industrial carpeting in two different phases. In the first phase, the subjects were asked to walk with their eyes open; in the second phase, the subjects were asked to perform the walking task with their eyes closed. The IMU sensor was mounted on the torso of each subject in order to collect gait data. From the raw sensor data, mean square values of resultant acceleration, angular velocity about the roll axis, angular velocity about the pitch axis, and angular velocity about the yaw axis were measured. To calculate the differences in the dependent measures, multilevel statistical techniques were used including separate models fitted to each dependent variable. Receiver operating characteristic (ROC) measures were taken into consideration for the statistical analysis of the sensor signal. Chi-square distribution was used to determine changes in eye open/closed conditions between pathological and non-pathological subjects. The experimental results in this study show that it is more likely for PN subjects to take significantly more consecutive steps with eyes open than eyes closed, while healthy individuals took more consecutive steps than PN subjects in both eyes open and closed conditions. The motion data analysis results indicated that PN subjects have higher angular velocity about the roll axis, angular velocity about the pitch axis, and angular velocity about the yaw axis. The results also showed that PN subjects show greater instability compared to healthy persons while performing the tandem task with their eyes closed.

Esser et al. [52] used a single IMU to analyze gait in order to diagnose peripheral neuropathy in diabetic patients. The main aim of the study was to analyze human gait using an IMU sensor during a 10-m walk test. The IMU was mounted on the lower back, and data from the accelerometer and gyroscope were recorded at the sampling rate of 100 Hz for further processing. By using the collected data, spatiotemporal gait parameters were extracted from the sensor data. The extracted parameters included step time, cadence, stride length, and walking speed. The data analysis for each group in order to find group differences was performed using the chi-square statistical method. This method compares the distribution of categorical variables in a sample with the distribution of categorical variables in a sample with the distribution of categorical variables in a sample with the distribution of categorical variables in a sample with the distribution of categorical variables in a sample with the distribution of categorical variables in a sample with the distribution of categorical variables in a sample with the distribution of categorical variables in a sample with the distribution of categorical variables in a sample with the distribution of categorical variables in another sample. For statistical analysis, this study uses ROC curves by means of the area under the curve (AUC). The experimental set-up included 17 participants (14 males, 3 females) with DPN, and 42 healthy participants (30 males, 12 females) aged around 63.2 ± 9.2 years. There was no difference in age, gender ratio, height, or BMI between groups. The participants were recruited from the Oxford Centre for Diabetes, Endocrinology and Metabolism at the Oxford University Hospitals NHS Foundation Trust

(Oxford, UK), and the study had approval from the National Research Ethics Committee (NRES: 11/SC/0218). The PN patients were confirmed using a monofilament test carried out by a specialist diabetes podiatrist. The obtained results in this study indicate that significant differences were found for all spatiotemporal parameters between PN patients and healthy subjects except for stride length. However, walking speed differed significantly in unhealthy and control subjects compared to any other gait-related parameters, while producing the largest discriminatory power (AUC = 0.975).

Wang et al. [53] proposed a framework for the diagnosis of peripheral neuropathy and other neurological disorders using two IMU sensors ((InvenSense MPU-6050) in order to analyze human gait. The two IMUs were used to measure five key kinematic and three spatiotemporal gait parameters that can help in distinguishing the type of neurological disorder. These parameters capture the kinematics of the ankle's linear motion and shank rotation, as dysfunction of lower-limb segments and joints would impact the motion of the ankle and shank. The IMUs were mounted on the ankle of each shank in the sagittal plane on the lateral side. The data were collected at a sampling frequency of 100 Hz. The study included 8 patients with PN (3 males, 5 females—age 49 + 8), 13 patients with post-stroke (PS) (9 males, 4 females—age 61 ± 15 years), 15 patients with Parkinson's disease (PD) (9 males, 6 females—age 76 \pm 7 years), and 13 healthy subjects (HC) (7 males, 6 femalesage 49 \pm 20 years). Additionally, information about subjects' heights and weights was provided in [53]. The study was approved by the Medical Ethics Committee of the School of Medicine at Zhejiang University, China. As can be observed, the data between each group are not balanced based on age or female/male ratio. In walking trials, the participants were asked to walk at a convenient speed for more than 12 m on a flat surface. From the IMU data, eight features were extracted based on detected gait phases and calculated motion trajectories i.e., stride length (SL), gait cycle duration (GD), percentage swing phase (PSP), max ankle velocity (MV), max ankle height (MH), ankle horizontal displacement (MHD), range of shank motion (RS), and kinematic asymmetry (KS). Using extracted features, a support vector machine (SVM) classifier was used to distinguish among four classes. The algorithm for training and validation of the proposed system was carried out in MATLAB. The classification accuracy achieved was 93.9% in this case. In this study, the authors used separate SVM classifiers for each of the four classes. In this way, four SVM classifiers were trained to distinguish data points of one class of subject from another class. During the training process, linear kernel function and sequential minimal optimization (SMO) method were used.

3.2. Pressure Sensor-Based Intelligent Wearable System for Diagnosis of PN

Foot plantar pressure is the distribution of the pressure field that acts between the foot and the surface [54]. It plays a very important role in the diagnosis of peripheral neuropathy. As in most cases, due to the damage of foot nerves which is common in diabetic patients, the patient cannot feel the right amount of plantar pressure required to walk smoothly. Due to nerve damage, the sensitivity of the foot decreases. Hence, patients having peripheral neuropathy in the lower limb will always exert more pressure than the healthy person while walking [55]. This makes it important for diabetic patients to have regular check-ups, as symptoms of PN will appear in later stages of the disease where it is not possible to recover the organ, and it may result in amputation of the foot [56]. By using a wearable pressure sensor, the early diagnosis of PN can be made possible, as wearable systems can be used in clinical settings as well as remotely [57]. Further, it is difficult for doctors to examine patients by simply observing the gait of the patients. Wearable sensors can provide an objective measurement and detailed knowledge about the physiology of the patient which is necessary to keep people's lives healthy [58]. There are a variety of pressure measurement systems available; however, they are broadly classified into two types: (1) platform systems and (2) in-shoe systems. This section of the paper summarizes the research related to recording and analyzing foot plantar pressure for the diagnosis of peripheral neuropathy.

Cao et al. [59] proposed a method for the diagnosis of PN based on foot plantar pressure distribution. The main aim of the study was to analyze foot plantar pressure changes while measuring foot plantar pressure distribution in order to prevent ulcers in elderly diabetic people. The study further investigates the role of plantar pressure in elderly diabetic patients with and without PN and compares pressure distribution between healthy subjects and diabetic patients with and without peripheral neuropathy. This study includes foot plantar pressure data from 19 diabetic patients with peripheral neuropathy (DPN) (10 males, 9 females—age 65.7 ± 2.4 years), 17 diabetic patients without peripheral neuropathy (D) (9 males, 8 females—age 65.2 \pm 6.8 years), and 20 healthy subjects (H) (11 males, 9 females—age 65.2 ± 5.4 years). Patients were recruited from Tianjin Medical University Chu Hsien-I Memorial Hospital in China. All subjects agreed to participate and signed informed consent after being fully informed of the study's procedure. The data were collected by using an insole wireless plantar pressure monitoring system designed by Medilogic, USA, at a sampling frequency of 300 Hz. The sensor can provide continuous data up to 10 m. The sensor was placed between the sole and the socks. The experimental protocol consisted of 10 m of walking trials at a speed suitable and comfortable for the subject. From plantar pressure data, the authors divided the plantar area into seven regions based on anatomical structure.

After collecting the plantar pressure data, the changes in peak pressures in segmented areas were analyzed while the subject was performing the walking task. In this study, the peak pressure was taken as the highest pressure in each segmented area during one gait cycle. A value of pressure higher than 200 KPa was considered high pressure. In this study, two insole pressure sensors were used to record pressure distribution in both feet. From sensor data, average pressure values of the right and left foot were calculated for each segment area in order to find statistical differences in each subject. The value of peak pressures in normal and DPN patients were investigated, clearly showing that DPN patients tend to have higher pressure in every segmented area of the foot compared to healthy or diabetic subjects without DPN. However, no differences were found in the peak pressure of healthy subjects and diabetic patients without PN. Results showed that the most sensitive areas related to the change in foot plantar pressure include the inner forefoot and medial forefoot region. Additionally, the peak pressure of the forefoot region in DPN patients is much higher compared to the peak pressure in the rear foot region. The overall study suggests that a significant increase in plantar pressure at the forefoot region was observed in DPN patients in the standing position, while healthy individuals in the standing position put more pressure on the rear foot.

Corpin et al. [60] proposed a model for the prediction of DPN by analyzing foot plantar pressure data. The Tekscan Medical Sensor 3000E hardware and Tekscan F-Scan 7.50 Research Software were used to collect the plantar pressure distribution of healthy subjects and DPN patients and to train different machine learning classifiers to distinguish between pathological and non-pathological subjects. The data collection step included 36 normal and diabetic volunteers and both female and male volunteers; however, the ratio between female and male volunteers was not provided. The volunteers' ages were between 49 and 56 years. The study was approved by the Ethics Committee under the supervision of the University of Santo Tomas Hospital (USTH), Manila, Philippines. The ground truth about the diagnosed case was obtained by using the Michigan Screening Instrument-questionnaire (MNSI-q). Nerve conduction velocity studies (NCV) were also conducted on each subject in order to find the ground truth. This research aims to classify among three classes: (1) healthy individuals (N); (2) individuals with diabetes but without peripheral neuropathy (DM); and (3) diabetic patients with peripheral neuropathy (DPN). Both male and female volunteers were included in these experimental trials. During walking trials, each subject had to walk 7 m in a straight line. Each subject had to perform the same walking trial eight times in order to obtain a sizable dataset. The subjects were asked to walk in their normal style during the trial. The Tekscan Medical Sensor 3000E hardware consists of 960 individual pressure sensing points to collect dynamic plantar

pressure data. The overall system was based on an in-shoe pressure measurement system. The output of each sensor point was divided into 256 increments to make the visualization better. The software divided the foot into 13 different regions by mapping the pressure data to accurately form the outline of the foot.

The parameters that were calculated by the software included peak pressure (PP), the instant of maximum force (IMxF), the instant of peak pressure (IPP), pressure–time integral (PTI), force-time integral (FTI), length of contact (LC), and contact area (CA). The data were then analyzed statistically using one-way ANOVA. Two different datasets were created for each leg separately, as some features were significantly different between the left and right foot. Principle component analysis (PCA) was also used to determine potential features and to remove those features that contributed less to the learning process by reducing the dimension of the dataset. From 208 features, PCA reduces the dimension of the dataset to only 29 new features that can successfully represent the overall data, i.e., 95%. Different machine learning classifiers such as SVM, random forest, multilayer perceptron (MLP), K-nearest neighbor (KNN), and Gaussian process (GP) were trained and tested in order to distinguish between three given classes. The k-fold cross-validation algorithm was also used to validate the system's performance. The results of this research showed that the parametric difference between the right and left foot is an indication of asymmetric plantar pressure distribution. Hence, separate datasets were used to compare parameters from both feet. The results also stated that instant maximum force time (IMxFT) and contact area (CA) on the right foot exhibit multiple significance in different regions. However, if there is a significant difference in the contact area between the left and right foot, it indicates the presence of DPN. For classification purposes, among the five classifiers, the SVM classifier outperforms other classifiers, with the highest accuracy of 91.91%.

Wang et al. [61] proposed a wireless footwear system to monitor diabetic foot ulcers due to peripheral neuropathy in diabetic patients. The system consists of an insole pressure sensor array that captures pressure changes during walking and transfers data via Bluetooth to a mobile phone in real time. So, the proposed methodology offers the continuous monitoring of plantar pressure using an insole flexible pressure system. A composite piezoresistive flexible sensor was developed to fulfill long-term monitoring requirements. The construction of the sensor was composed of carbon black and silicon rubber. The functioning of the sensor was validated by a custom pressure testing platform, which consists of a keyboard testing machine, a pressure testing machine, and a desktop computer. The data of 5 healthy subjects (HC) (4 males, 1 female—age 48.5 \pm 3.5 years), 5 diabetic patients without PN (D) (2 males, 3 females—age 55.8 \pm 5.6 years), and 5 DPN patients (2 males, 3 females—age 59.00 \pm 10.71 years) were recorded by the proposed pressure measurement system. All subjects were able to walk without any gait aids. The study was approved by the Research Ethics Committee of the Body Data Science Engineering Center of Guangdong Province and the First Affiliated Hospital of Jinan University of Guangdong Province in China. In the walking trial, subjects were asked to walk 20 m along the corridor and stairs for one minute at their regular speed. The data were captured at a sampling frequency of 20 Hz with a 12-bit sampling resolution. The application interface for smartphones was also designed to allow users to visualize the pressure distribution in real time. The final database consists of 2403 samples, including 779 samples from the HC group, 736 from the D group, and 888 samples recorded for DPN patients. From the raw sensor data, peak plantar pressure (PPP), the pressure-time integral (PTI), the maximum pressure gradient (MaxPG), the minimum pressure gradient (MinPG), the full width at half maximum (FWHM), the forefoot-to-rearfoot plantar pressure ratio (F/R), and the symmetry index (SI) were extracted for the database feature points. The feature points used in this study were calculated using the method given by Botros et al. [62]. Five different machine learning classifiers, support vector machine (SVM), K-nearest neighbors (KNN), RF (random forest), GBDT (gradient-boosted decision trees), and AdaBoost classifiers were trained, and they tested the performance of the proposed system. A 10-fold cross-validation was also used to validate the ML models. The overall average accuracy of all five classifiers

used was 85%, with the highest accuracy of 94.7% in the case of the random forest (RF) classifier. It should be noted that the sample size of the participants (five per class) is a small number, and further evaluation is needed with a higher number of participants.

3.3. ECG-Based Intelligent Wearable Systems for Diagnosis of PN

According to [63], almost one-third of acute myocardial infarction patients have diabetes, which is one of the leading causes of peripheral neuropathy. The damage to these peripheral nerves that regulate the heart mechanism is called cardiac autonomic neuropathy (CAN) [64]. Electrocardiography (ECG) is a quick and non-invasive procedure for the early detection of CAN, which is caused by the damage of those peripheral nerves that are responsible for the proper functioning of the heart. This section of the paper focuses on the diagnosis of CAN by analyzing ECG signals.

In Ref. [65], an ECG-based platform for the diagnosis of peripheral neuropathy was proposed. The diagnosis was based on the fact that uncontrolled high glucose levels in diabetic patients cause cardiovascular diseases because they affect heart rate variability (HRV). This study is focused on analyzing HRV parameters in DPN patients so that normal and abnormal ECG can be distinguished. The HRV analysis has the capability to recognize variations in the autonomic nervous system (ANS), which is responsible for keeping the heart functioning properly [66]. The data collection method involved the recording of an ECG for all participants for 24 h using a four-channel Holter machine. Twenty subjects were enrolled for the data collection task. All 20 subjects (10 males and 10 females with a mean age of 55.7 years old) were type-II diabetes mellitus patients and were all 40 years old or above. Of the 20 subjects, 10 subjects were confirmed diagnosed with DPN using nerve conduction studies (NCS). The study was approved by the ethical committee of Bangladesh University of Health Sciences. The ECG data were collected at a 200 Hz sampling rate and then the signal was processed in order to remove any possible noise. The ECG signal was then processed in order to extract nine feature points related to HRV to distinguish between DPN-positive and DPN-negative groups. Both time-domain and frequency-domain parameters of the ECG signal were extracted for classification purposes. The Wilcoxon rank-sum test [67] was used to find statistical significance between these two classes. The obtained results showed the feasibility of ECG data for the purpose of diagnosing peripheral neuropathy related to the heart.

In Ref. [68], Jelinek et al. also used HRV attributes of the ECG signal in order to distinguish between the DPN group and the healthy control group. This study proposed a new classification technique for diagnosis purposes and compared the performance of other machine learning (ML) classifiers with the proposed one. The authors utilized their prior collected dataset called Diab Health [69] and selected a subset of 21 patients with severe diabetic neuropathy. However, no information was provided on how the subset was selected. The main aim of the study was to investigate the contribution of HRV parameters in an automated disease classification task. A multi-level clustering technique was used to improve diagnostic accuracy. The data collection procedure included 20 min of ECG recording in a supine position for all participants. The ECG signals were recorded at a sampling frequency of 400 Hz with a lead II configuration [70]. The ground truth for CAN patients was acquired from Ewing battery criteria [71]. The results of this study indicate the significance of HRV parameters for the diagnosis of cardiovascular diseases and the proposed graph-based machine learning classification algorithm (GBML) performed better compared to other conventional clustering techniques. The performances of ML algorithms were based on sensitivity and specificity, which are common metrics for ML algorithms [72]. The best sensitivity of 0.98 and the best specificity of 0.89 were achieved in the case of the GBML clustering technique.

Sharanya and Sridhar [73] proposed a system for the diagnosis of CAN that is based on a convolutional neural network (CNN) for prediction purposes. The aim of this study was to classify CAN-positive and CAN-negative subjects by analyzing ECG signals. In the experimental trials, 13 male and 6 female subjects participated. Among them, 9 subjects were confirmed positive for CAN, and 10 were healthy subjects, labeled as CAN-negative class. No information on the age of the participants or approval of the study from an institutional review board or an ethical committee was provided. The ECG signals were acquired at a sampling frequency of 400 Hz in lead II configuration for 20 min. After acquiring the ECG signal, the signal was processed to remove possible noise, and then feature points such as RR-intervals, etc., were extracted from the clean ECG signal. Using the extracted features, a CNN model was trained to distinguish between normal ECG and abnormal ECG, representing the presence of CAN. The overall results show that CNN can take care of analyzing the complexity of the ECG attributes for the early diagnosis of CAN. The overall accuracy for the diagnosis of CAN achieved in this study is 95.42%. Research work discussed in this paper for the diagnosis purpose using a single wearable sensor is summarized in Table 1.

Table 1. Summary of the wearable systems for detecting peripheral neuropathy (PN) using a single-type sensor. H: healthy; DPN: diabetic PN; CAN: cardiac autonomic neuropathy; PD: Parkinson's disease; ACC: accelerometer; IMU: inertial measurement unit; N/A: not available.

References	No. of Participants	Distribution of Participants	No. of Sensors and Placement	Data Collection Procedure	Methodology	Results
Chen and Shanshan [50]	106	PN: 30 H: 76	3-axis ACC	Single ear-worn ACC	10-m walking test	Gait analysis using logistic regression models for training and testing
Cohen et al. [51]	72	PN: 21 H: 61	IMU	Single torso-mounted IMU	Tandem walking test	ROC and Chi-square methods were used to evaluate the gait data.
Esser et al. [52]	56	PN:14 H: 42	IMU	Single IMU attached to lower back	Standard 10-m walking test	Chi-square distribution using IMU Sensor data for classification. Statistical analysis was conducted using ROC.
Wang et al. [53]	49	PN: 9 Stroke: 13 PD: 14 H: 13	IMUs	Two IMUs were attached to the ankle of each shank	12-m walking trail	The gait parameters were extracted using method [74] based on wavelet analysis.
Cao et al. [59]	56	DPN: 19 Diabetic: 17 H: 20	Insole wireless plantar pressure monitoring system designed by Medilogic	The sensor was placed on one foot between the soles and socks of the participants	10-m walking test recorded at 300 Hz sampling	Peak pressure was recorded in each case by dividing the foot into seven segments and then comparing the pressure distribution of each region in each of the two classes
Corpin and Ryan Rey A. [60]	36	N/A	Tekscan Medical Sensor 3000E	Single Tekscan Medical sensor placed on the right foot only.	7 m walking in a straight line and repeat the procedure eight times	In-shoe pressure monitoring system was used. The Tekscan software provides a number of gait and pressure parameters that can be used as features for ML algorithms
Wang et al. [61]	20	DPN: 5 H: 5	Insole piezoresistive pressure sensor array	Two insole pressure sensors that each contained eight pressure measuring points were placed on both feet.	20-m walking test	Using the proposed insole system, the pressure data were collected from each sensing point, and peak pressures were recorded to create a database of healthy and unhealthy subjects. Five different classification algorithms were then trained for th diagnosis, and the model was validated by using k-fold validatior
Morshed et al. [65]	20	DPN: 10 H: 10	Holter device	Four-channel (RA-LA, LA-LL, LL-RA, and Vx-RL) Holter device	24-h ECG recording at 200 Hz	HRV parameters were extracted from ECG data using a method in [75]. Both time-domain and frequency-domain features of the ECG signal were used in the diagnosis of PN.
Jelinek et al. [68]	21	DPN: 21	ECG	ECG sensor with lead II configuration	20-min ECG recording in spine position	Using HRV attributes of the ECG signal, a new multi-level clustering technique was proposed and implemented to distinguish betwee two classes.
Sharanya, S. and P.A. Sridhar [73]	19	CAN: 9 H: 10	ECG	ECG sensor with lead II configuration	20-min ECG recording	A CNN network was used to distinguish between PN and health subjects. A 20-min-long ECG was recorded for each subject.

4. Intelligent Wearable Multisensory Systems for Diagnosis of PN

Due to their multiple benefits over conventional devices, wearable sensors have gained tremendous interest in the clinical and medical fields [76]. The main features of wearable sensors are their size and flexibility. Due to their small size, multiple wearable sensors can be worn by the subject without feeling bulky. This provides continuous real-time data and simultaneously offers a way to collect data from multiple body parts [77]. This section describes the research work that is based on multiple sensors attached to different locations on the human body in order to diagnose peripheral neuropathy.

Sejdic et al. [78] used accelerometer data to model the human gait by extracting gait parameters from the raw accelerometer signal and showed the effectiveness of gait analysis by extracting clinically valuable information from a three-axis accelerometer signal for the diagnosis of peripheral neuropathy (PN) and Parkinson's disease (PD). The accelerometer was attached to the torso of the human subjects. The placement of the sensor was at the torso due to the sensitivity of torso dynamics to age and disease-related gait changes. The main purpose of the study was to comprehensively examine multiple gait features provided by the accelerometer signal in multiple domains across healthy and unhealthy subjects. The features from time, frequency, and time-frequency domain were extracted in order to distinguish among three considered groups, i.e., subjects with PN or PD or healthy subjects. The experimental procedure included a walking trial of subjects aged 65 years or above on a custom computer-controlled treadmill. Among the participants, there were 14 healthy subjects, 11 subjects with peripheral neuropathy, and 10 subjects with Parkinson's disease. The authors did not provide information about the gender of the participants. The study was approved by the Institutional Review Board at the University of Pittsburgh, USA. Dynamic and static reflective biomarkers were also placed on the bony landmarks in order to measure heel and toe trajectory data for stride segmentation using a 3D optical motion capture system (made by Natural Point, Inc., Corvallis, OR, USA). A three-axis accelerometer (MMA7260Q, Freescale Semiconductor, Austin, TX, USA) was firmly attached over the L3 segment of the lumbar spine to measure linear acceleration along vertical, anterior-posterior, and medial-lateral axes. In each waking trial, the data were recorded at 100 Hz for 3 min after the subject reached the optimal or preferable speed. From the toe and heel trajectory data, strides were calculated by using 3D optical motion capture system data. The system used coordinates of the heel and the toe markers in the direction of progression in order to calculate the actual instantaneous speed and strides, a method proposed by Fusco, N. and Cretual, A. [79]. From the stride data, gait speed, mean stride intervals, and coefficient of variations were calculated. From the acceleration signal, statistical features such as standard deviation, skewness, and kurtosis were calculated. Different frequency domain parameters were also extracted, such as cross-entropy, centroid frequency, bandwidth, wavelet bands, and wavelet entropy. The overall results showed that a person with peripheral neuropathy walks more slowly compared to healthy individuals. The features extracted from the acceleration signal proved to be quite suitable for detecting peripheral neuropathy and other neurological disorders.

Khandakar et al. [80] proposed a technique to diagnose foot ulcers in diabetic patients caused by peripheral neuropathy due to excessive amounts of glucose in the blood. The main technique in their study was based on monitoring temperature and pressure changes in the diabetic foot in order to detect any chances of a foot ulcer due to PN. Smart insole pressure sensor (force-sensitive resistor) and temperature sensors (flexible thermistor) were used to record data from both feet. The proposed system provides portable and wireless transmission of pressure and temperature sensor data and can be used for long-term monitoring of the foot. Each insole sensor for each foot contains sixteen force-resistive pressure sensors and eight temperature sensors. In the walking experiment, each subject was asked to perform a 20 m walk six times, and data were collected at the sampling frequency of 40 Hz. The total number of participants was 12, aged between 20 and 59 years and included both female and male participants; however, the gender ratio was not defined. The local ethical committee of Qatar University approved the study. After that, each gait

cycle was segmented in order to obtain the mean and standard deviation of the segmented gait cycles. The overall results show that the proposed system combined with ML provides a low-cost solution to diagnose or monitor diabetic foot ulcers due to PN. It also offers a way to transmit pressure and temperature data wirelessly at a low cost and with an appropriate number of sensors.

Another similar approach for the diagnosis of peripheral neuropathy in the foot or foot ulcers was given by [81], which was also based on measuring foot temperature and pressure in order to find the presence of PN. However, the proposed system is IoT-based open source. The study was based on the fact that DPN patients usually lose sensations in their feet during walking and standing due to damage to peripheral nerves and have higher foot temperatures than healthy individuals. The proposed systems offer cloud access to real-time pressure and temperature data. The severity of DPN was measured by analyzing higher pressure and higher temperature values over multiple positions on the foot. The system consisted of a flat FlexiForce pressure sensor embedded in a shoe sole, and a couple of DHT11 temperature sensors were attached to record pressure distribution. The overall hardware set-up for data collection during walking trials and pressure and temperature values used in this research was based on a theoretical analysis of the foot representing different levels of severity of the disease. The proposed research showed how cloud-based monitoring of DPN patients can help diabetes patients stay healthy by avoiding ulcer that are caused by PN. The system can monitor patients for a very long time and is capable of sending alert warnings if the data show some abnormal signs or patterns.

Sempere-Bigorra et al. [82] also presented a technique to monitor and diagnose diabetic foot ulcers due to PN using a wearable inertial sensor. A single IMU in the lumbar region was used to model the gait; however, this research used additional data acquisition and diagnostic methods such as sensory and vibration tests to diagnose PN. Hence, the scope of this research falls into the multisensory system category. Spatiotemporal gait parameters were extracted from the IMU signal and the superficial sensitivity and vibration test. Superficial sensitivity was performed using nociception tests [83], and deep sensitivity was examined using vibration tests [84]. The similarities between gait parameters and each sensory test were analyzed using a logistic regression model to find the presence of PN in the subjects. In the walking experiment, each subject was asked to walk 15 m back and forth at their normal speed. A total of 85 subjects (46 males and 39 females) participated in the experiment. All participants were aged between 20 and 87 years with a mean value of 68.1 ± 1.3 years. The study was approved by the University of Bologna Ethics Committee for Human Research in Italy. The IMU used in this research was manufactured by Wiva Science and is capable of transmitting data wirelessly via Bluetooth. For data distribution analysis, the Kolmogorov–Smirnov test was used [85], which showed nonnormal distribution among data from different tests. Logistic regression analysis of the data showed that the inertial data and other clinical parameters can be used to successfully diagnose PN, as they are associated with lower sensitivity which shows the signs of PN in its early stages. The main outcome of the study was to measure the relationship between sensitivity impairment and feasible gait parameters. The observed relationship showed that it can be used in non-laboratory settings for the diagnosis of peripheral neuropathy and that IMUs provide valuable information in the diagnosis process.

Z. Veličković et al. [86] presented an approach for the early detection of PN caused by systemic autoimmune rheumatic diseases (SARDs). The authors focused on the use of wearable sensors along with machine learning as a screening tool for the diagnosis of PN related to SARDs. In the experimental study, both healthy and SARD patients participated. The total number of participants was 23 (9 males and 14 females), among which there were 11 SARD patients, and the rest were healthy subjects. The experiment consisted of recordings of different types of data. First, nerve conduction study (NCS) results were obtained for each subject. The results of the NCS studies provided the conduction velocity of motor and sensory fibers. In the second stage, four IMUs attached to both legs and arms were used to model the motion of subjects while performing six different types of exercises with eyes closed and open. The raw IMU data were sent to a tablet through Bluetooth, which sent it to the central server through Wi-Fi for feature extraction such as acceleration and power. The proposed system used a binary machine learning classifier for the detection of the presence or non-presence of PN-related SARD. The performance of the proposed model showed fine sensitivity and specificity. The results of the study highlight the potential of using wearable technology for the accurate diagnosis of diseases related to peripheral neuropathy.

Table 2 shows the summary of research work using multiple sensors.

Table 2. Summary of the wearable systems for detecting peripheral neuropathy (PN) using multisensory systems.

References	No. of Participants	Distribution of Participants	No. of Sensors and Placement	Data Collection Procedure	Methodology	Results
Sejdic et al. [78]	35	DPN: 11 PD: 10 H: 14	Accelerometer (ACC) and 3D optical motion capture system (Natural Point, Inc., Corvallis, OR, USA)	Single ACC attached at the torso	3 m walking test	Accelerometer and 3D optical system captured the gait data at 100 Hz. Different spatiotemporal and frequency domain features were extracted.
Khandakar et al. [80]	12	N/A	Force-sensitive resistors and temperature sensors based on thermistor	Two in-shoe wireless pressure and temperature monitoring systems using 16 FSR and 8 temperature sensors for each foot.	20 m walking test at a sampling rate of 40 Hz	NodeMCU and multiplexer were used to send all the data wirelessly to the main computer.
Kukreja et al. [79,81]	N/A	N/A	FlexForce Sensor and two DHT11 temperature sensors	In-shoe flexi pressure sensor was placed in the sole of the shoe, and two temperature sensors were aligned parallel to the instep and sole.	The data were collected from the participant wirelessly using NodeMCU and in-shoe sensors	Threshold values for the discrimination between healthy and PN patients were evaluated and used to diagnose PN
Sempere- Bigorra et al. [82]		DPN: 8 Diabetic: 77	IMU, vibration test, and sensitivity test	IMU attached to a lumbar area on the L5 spinal segment	The gait data were collected by using an IMU sensor, and data from other tests such as vibratory and sensitivity tests were also collected	The data were analyzed in order to find a correlation between these data. Logistic regression was then used to find the similarities between different groups.
Z. Veličković et al. [86]		PN: 11 Healthy: 12	NCS and four IMUs	NCS electrodes were placed at the chest, and IMUs were attached to each leg and arm	First, NCS data were collected for all participants, and then six different exercises were performed to record and process IMU data	The obtained results show that wearable devices can be made useful in the diagnosis of PN. The overall results showed good specificity and sensitivity.

5. Discussion

In peripheral neuropathy, the gait of the patient is affected mostly due to pain, numbness, etc. Hence, all the studies given in this paper, except for the ones based on ECG, are based on the analysis of the human gait and foot plantar pressure for the diagnosis of PN in lower region muscles. Research studies [50-52] include the usage of a single inertial sensor placed at different body locations, such as the ear or leg in order to model the human gait. Each study had its own experimental protocol and extracted information that was relevant to the diagnosis. These studies show that an IMU or accelerometer may play an important role in the early diagnosis of PN, which is very crucial for human well-being. Research studies such as [53] show that the human gait can be modeled more accurately with the use of two inertial sensors attached to both legs. These studies also show how important the placement and number of sensors are to model gait exactly for such a task. Another important parameter in the diagnosis of PN is foot plantar pressure. To measure foot plantar pressure, mostly in-shoe-based systems are used. Research studies [59–61] discussed in this paper are all based on in-shoe pressure-measuring systems. However, these studies show the importance of measuring foot plantar pressure with the correct number of pressure points and locations. The gait parameters obtained from the in-shoe

systems in all research studies were different, which shows the importance of recorded features for capturing gait using foot plantar pressure.

The research studies on the diagnosis of cardiac autonomic peripheral neuropathy (CAN) are also included along with gait analysis-based systems. The reason for this is that a strong connection or relationship can be modeled between the gait and ECG of the patient since the ECG patterns change during a walk. These patterns will differ in healthy persons compared to patients having PN at lower legs or CAN. The discussed research in this paper on the diagnosis of CAN is mostly based on analyzing heart rate variability which gives proper measures to diagnose CAN. The main difference in CAN-related studies is the use of different classifiers for the classification of disease.

It is also important to note that there is no universal protocol for data collection for PN. Therefore, each study defined its own experimental protocol for PN detection based on widely used methods by physicians. Therefore, the studies cannot be easily compared, as their protocols, heterogeneity in subjects, and data collection process were different. Additionally, since different sensors or sensor types are used in each study, there are differences in data, such as frequency, sample rate, range, and others, which contribute to the challenge of developing a benchmark dataset for PN detection.

6. Open Challenges and Conclusions

Recently, the usage of wearable devices has been increasing in many fields including healthcare. The prevalence of diabetes across the world contributes the most to causing peripheral neuropathy. Due to the increasing number of people suffering from peripheral neuropathy, there is a need to implement efficient methods based on modern technology and machine learning. In this section, the open challenges are summarized.

Readiness of wearable technology: Wearable devices can play a vital role in the diagnosis of many diseases, as they can be worn easily, multiple sensors can be used simultaneously, and continuous sources of data from the patient can be recorded for the investigations [87]. However, wearable systems are currently not that mature when it comes to the diagnosis of neurological diseases in clinical settings such as peripheral neuropathy [88]. There are a few commercially available wearable devices for healthcare purposes, but none of the wearable device systems has been used successfully for the accurate and early diagnosis of PN in clinical settings.

Easy-to-use systems for doctors and/or patients: Wearable sensor systems that are discussed in this paper are usually set up by engineers and researchers. To ensure that these systems are widely used, they need to take into account how doctors and/or patients (end-users) would use the systems. For example, a doctor should be able to set up the system, connect with the sensors, and understand what the data show (e.g., through a graphical user interface). Similarly, a patient should be able to wear the system at home and send the data to their doctor if remote and/or continuous monitoring is needed. Improving user experience is a critical factor for both patients and doctors.

Privacy concerns: Data privacy is an important issue when a system collects human data. Modern wearable technology has enabled us to combine information technology such as cloud medicine, big data, etc., to help monitor patients, but, at the same time, it also poses security threats since the data can be hacked and misused. Furthermore, companies can gather and trade data from smartphones or wearables to obtain potentially important information about users for different purposes, which compromises user privacy [89].

Ethical Considerations: Wearable devices provide new and innovative ways to help in the field of healthcare; however, certain ethical considerations are needed regarding the use of wearables in healthcare especially for third-party consumer tech. Ethical considerations should be clearly defined for how the data will be collected and what protocols have to be followed for the respective purpose so that companies will not misuse the data for their profits [90]. Additionally, data bias should also be addressed. In the papers presented in this survey, it is clear that very few studies had balanced datasets based on age or gender. For the US-based studies, race was not reported. Having datasets that are not distributed based on age, gender, race, etc., has the potential of creating ML models that are biased. AI models may increase mistrust and health inequality [91]. Explainable AI has the potential to address some of the biases [92,93]; however, it is important to have policies that would enable unbiased datasets [91].

User Awareness: Due to the relatively new technology, most users do not know the importance of the privacy of their data which are collected by wearable devices, especially in healthcare applications. Users need to understand the risks associated with wearable technology. In the future, this issue must be investigated extensively in order to ensure user privacy [93].

Accuracy and Reusability: Accuracy is one the most important factors when it comes to wearable systems for healthcare. If the accuracy of a wearable device is not sufficient, then this may lead to an incorrect diagnosis since the collected data will miss important vital signs. Another important factor is the reusability of the device. It is important to design wearable systems in such a way that they can be used continuously, and data must be collected with high precision. These two factors play a very important role in making wearable devices more suitable for healthcare [94].

Placement and Optimal Number of Wearable Devices: The correct placement of the wearable device on the human body is very necessary to collect accurate information about the user. The placement of wearable sensors affects accuracy, capability, and user experience. It is also important to use the right type of wearable device depending on the location of the body at which the system will be worn and the nature of the data [95]. It is also crucial to optimize the number of wearable devices that are needed and to define placement locations that are easily accessible. If a system has several devices and is complex to be worn, the physicians and/or the patients may be reluctant to use it. Recent work by Ha et al. [96] conducted a study to determine the optimal position for a pair of electrocardiography (ECG) and thermocouple (TC) sensors (for body temperature measurement) and a pair of photoplethysmography (PPG) and TC sensors. Therefore, additional research is required for the identification of optimal wearable devices, such as biosensors and motion sensors, and their placement while maintaining a high level of accuracy of PN detection.

Wearable devices are required to be at a high technology readiness level [97], but legislation and policy are also required to provide protections for human data [98]. Therefore, it is clear that more research is needed in the field of wearable devices to address these challenges so that they can be used effectively in healthcare [99], and policies are required to protect patients and to ensure trustworthy and unbiased systems.

Author Contributions: Conceptualization, M.T., M.K. and E.A.B.; materials related to wearable systems detecting peripheral neuropathy, M.T. and M.K.; investigation of the open challenges, M.K. and M.T.; writing—original draft preparation, M.T. and M.K.; writing—review and editing, M.T., M.K. and E.A.B.; Project administration, E.A.B. and M.K. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Hughes, R.A. Peripheral neuropathy. BMJ 2002, 324, 466–469. [CrossRef] [PubMed]
- Stino, A.M.; Smith, A.G. Peripheral neuropathy in prediabetes and the metabolic syndrome. J. Diabetes Investig. 2017, 8, 646–655. [CrossRef]
- Staff, N.P.; Grisold, A.; Grisold, W.; Windebank, A.J. Chemotherapy-induced peripheral neuropathy: A current review. *Ann. Neurol.* 2017, *81*, 772–781. [CrossRef] [PubMed]

- Castelli, G.; Desai, K.M.; Cantone, R.E. Peripheral Neuropathy: Evaluation and Differential Diagnosis. *Am. Fam. Physician* 2020, 102, 732–739. [PubMed]
- 5. Anastasi, J.K.; Capili, B. Detecting Peripheral Neuropathy in Patients with Diabetes, Prediabetes and other High-Risk Conditions: An Advanced Practice Nurse's Perspective. *J. Med. Clin. Nurs.* **2022**, *3*, 143. [CrossRef]
- 6. Yorek, M.; Malik, R.A.; Calcutt, N.A.; Vinik, A.; Yagihashi, S. Diabetic Neuropathy: New Insights to Early Diagnosis and Treatments. *J. Diabetes Res.* **2018**, 2018, 5378439. [CrossRef]
- Watson, J.C.; Dyck, P.J. Peripheral Neuropathy: A Practical Approach to Diagnosis and Symptom Management. *Mayo Clin. Proc.* 2015, 90, 940–951. [CrossRef]
- 8. Burgess, J.; Frank, B.; Marshall, A.; Khalil, R.S.; Ponirakis, G.; Petropoulos, I.N.; Cuthbertson, D.J.; Malik, R.A.; Alam, U. Early Detection of Diabetic Peripheral Neuropathy: A Focus on Small Nerve Fibres. *Diagnostics* **2021**, *11*, 165. [CrossRef]
- Del Core, M.A.; Ahn, J.; Lewis, R.B., III; Raspovic, K.M.; Lalli, T.A.; Wukich, D.K. The Evaluation and Treatment of Diabetic Foot Ulcers and Diabetic Foot Infections. *Foot Ankle Orthop.* 2018, *3*, 2473011418788864. [CrossRef]
- Selvarajah, D.; Kar, D.; Khunti, K.; Davies, M.J.; Scott, A.R.; Walker, J.; Tesfaye, S. Diabetic peripheral neuropathy: Advances in diagnosis and strategies for screening and early intervention. *Lancet Diabetes Endocrinol.* 2019, 7, 938–948. [CrossRef]
- Park, E.; Kim, K.J.; Kwon, S.J. Understanding the emergence of wearable devices as next-generation tools for health communication. *Inf. Technol. People* 2016, 29, 717–732. [CrossRef]
- 12. Mukhopadhyay, S.C.; Suryadevara, N.K.; Nag, A. Wearable Sensors for Healthcare: Fabrication to Application. *Sensors* 2022, 22, 5137. [CrossRef] [PubMed]
- 13. Amft, O.; Lukowicz, P. From Backpacks to Smartphones: Past, Present, and Future of Wearable Computers. *IEEE Pervasive Comput.* 2009, *8*, 8–13. [CrossRef]
- 14. Yang, M.; Ye, Z.; Ren, Y.; Farhat, M.; Chen, P.-Y. Recent Advances in Nanomaterials Used for Wearable Electronics. *Micromachines* **2023**, *14*, 603. [CrossRef] [PubMed]
- Huhn, S.; Axt, M.; Gunga, H.-C.; Maggioni, M.A.; Munga, S.; Obor, D.; Sié, A.; Boudo, V.; Bunker, A.; Sauerborn, R.; et al. The Impact of Wearable Technologies in Health Research: Scoping Review. *JMIR Mhealth Uhealth* 2022, 10, e34384. [CrossRef]
- 16. Almusawi, H.A.; Durugbo, C.M.; Bugawa, A.M. Wearable Technology in Education: A Systematic Review. *IEEE Trans. Learn. Technol.* **2021**, *14*, 540–554. [CrossRef]
- 17. Li, R.T.; Kling, S.R.; Salata, M.J.; Cupp, S.A.; Sheehan, J.; Voos, J.E. Wearable Performance Devices in Sports Medicine. *Sports Health* **2016**, *8*, 74–78. [CrossRef]
- Kodam, S.; Bharathgoud, N.; Ramachandran, B. A review on smart wearable devices for soldier safety during battlefield using WSN technology. *Mater. Today Proc.* 2020, 33, 4578–4585. [CrossRef]
- Veenstra, B.; Friedl, K. Military applications of wearable physiological monitoring—From concept to implementation. J. Sci. Med. Sport 2017, 20, S133. [CrossRef]
- Ometov, A.; Shubina, V.; Klus, L.; Skibińska, J.; Saafi, S.; Pascacio, P.; Flueratoru, L.; Gaibor, D.Q.; Chukhno, N.; Chukhno, O.; et al. A Survey on Wearable Technology: History, State-of-the-Art and Current Challenges. *Comput. Netw.* 2021, 193, 108074. [CrossRef]
- Aroganam, G.; Manivannan, N.; Harrison, D. Review on Wearable Technology Sensors Used in Consumer Sport Applications. Sensors 2019, 19, 1983. [CrossRef] [PubMed]
- Fuller, D.; Colwell, E.; Low, J.; Orychock, K.; Tobin, M.A.; Simango, B.; Buote, R.; Van Heerden, D.; Luan, H.; Cullen, K.; et al. Reliability and Validity of Commercially Available Wearable Devices for Measuring Steps, Energy Expenditure, and Heart Rate: Systematic Review. *JMIR Mhealth Uhealth* 2020, *8*, e18694. [CrossRef] [PubMed]
- 23. Wen, D.; Zhang, X.; Liu, X.; Lei, J.; Mira, J.J.; Fernández, C. Evaluating the Consistency of Current Mainstream Wearable Devices in Health Monitoring: A Comparison Under Free-Living Conditions. J. Med. Internet Res. 2017, 19, e68. [CrossRef] [PubMed]
- 24. Lu, L.; Zhang, J.; Xie, Y.; Gao, F.; Xu, S.; Wu, X.; Ye, Z. Wearable Health Devices in Health Care: Narrative Systematic Review. *JMIR Mhealth Uhealth* **2020**, *8*, e18907. [CrossRef]
- 25. Barriga, E.S.; Chekh, V.; Carranza, C.; Burge, M.R.; Edwards, A.; McGrew, E.; Zamora, G.; Soliz, P. Computational basis for risk stratification of peripheral neuropathy from thermal imaging. In Proceedings of the 2012 Annual International Conference of the IEEE Engineering in Medicine and Biology Society, San Diego, CA, USA, 28 August–1 September 2012; pp. 1486–1489.
- 26. Gallardo, E.; Noto, Y.-I.; Simon, N.G. Ultrasound in the diagnosis of peripheral neuropathy: Structure meets function in the neuromuscular clinic. *J. Neurol. Neurosurg. Psychiatry* **2015**, *86*, 1066–1074. [CrossRef]
- Agurto, C.; Chek, V.; Edwards, A.; Jarry, Z.; Barriga, S.; Simon, J.; Soliz, P. A thermoregulation model to detect diabetic peripheral neuropathy. In Proceedings of the 2016 IEEE Southwest Symposium on Image Analysis and Interpretation (SSIAI), Santa Fe, NM, USA, 6–8 March 2016; pp. 13–16.
- Ferreira, A.; Lamas, J.; Gomes, L.; Silva, S.; Loureiro, C.; Domingues, J.P.; Silva, J.S.; Morgado, M. neuroCornea—Diabetic peripheral neuropathy early diagnosis and follow-up through in vivo automatic analysis of corneal nerves morphology. In Proceedings of the 1st Portuguese Biomedical Engineering Meeting, Lisbon, Portugal, 1–4 March 2011; pp. 1–4.
- Sawada, H.; Uchida, K.; Danjo, J.; Nakamura, Y. Development of a non-invasive screening device of diabetic peripheral neuropathy based on the perception of micro-vibration. In Proceedings of the 2016 IEEE Conference on Computational Intelligence in Bioinformatics and Computational Biology (CIBCB), Chiang Mai, Thailand, 5–7 October 2016; pp. 1–6.

- Silva, S.F.; Loureiro, C.F.; Almeida, H.; Otel, I.; Domingues, J.P.; Silva, J.S.; Quadrado, M.J.; Morgado, A.M. Evaluation of corneal nerves morphology for diabetic peripheral neuropathy assessment. In Proceedings of the 2012 IEEE 2nd Portuguese Meeting in Bioengineering (ENBENG), Coimbra, Portugal, 23–25 February 2012; pp. 1–4.
- Barthakur, M.; Hazarika, A.; Bhuyan, M. Rule based fuzzy approach for peripheral motor neuropathy (PMN) diagnosis based on NCS data. In Proceedings of the International Conference on Recent Advances and Innovations in Engineering (ICRAIE-2014), Jaipur, India, 9–11 May 2014; pp. 1–9.
- D'Angelo, M.L.; Cannella, F.; Liberini, P.; Caldwell, D.G. Development and validation of a tactile sensitivity scale for peripheral neuropathy screening. In Proceedings of the 2014 36th Annual International Conference of the IEEE Engineering in Medicine and Biology Society, Chicago, IL, USA, 26–30 August 2014; pp. 4823–4826.
- Komori, H.; Watanabe, K.; Tsuichihara, S.; Takemura, H.; Imai, M.; Haraguchi, M.; Chou, S. Screening System for Diabetes Peripheral Neuropathy Using Foot Plantar Images on Different Hardness Floor. In Proceedings of the 2019 IEEE International Conference on Systems, Man and Cybernetics (SMC), Bari, Italy, 6–9 October 2019; pp. 2614–2619.
- Benson, J.; Estrada, T.; Burge, M.; Soliz, P. Diabetic Peripheral Neuropathy Risk Assessment using Digital Fundus Photographs and Machine Learning. In Proceedings of the 2020 42nd Annual International Conference of the IEEE Engineering in Medicine & Biology Society (EMBC), Montreal, QC, Canada, 27 August 2020; pp. 1988–1991.
- Kim, W.; Kim, Y. Human Body Model using Multiple Depth Camera for Gait Analysis. In Proceedings of the 2018 19th IEEE/ACIS International Conference on Software Engineering, Artificial Intelligence, Networking and Parallel/Distributed Computing (SNPD), Busan, Republic of Korea, 27–29 June 2018; pp. 70–75.
- Yu-Ren, L.; Miaou, S.G.; Hung, C.K.; Sese, J.T. A gait analysis system using two cameras with orthogonal view. In Proceedings of the 2011 International Conference on Multimedia Technology, Hangzhou, China, 26–28 July 2011; pp. 2841–2844.
- Chikano, M.; Konno, T.; Awai, S. Robust Gait Recognition for Occlusion Caused by Surveillance Cameras. In Proceedings of the 2021 IEEE 10th Global Conference on Consumer Electronics (GCCE), Kyoto, Japan, 12–15 October 2021; pp. 148–149.
- Ferreira, J.P.; Liu, T.; Coimbra, P.; Coimbra, P. Parameter analysis and selection for human gait characterization using a low cost vision system. In Proceedings of the 2017 4th International Conference on Systems and Informatics (ICSAI), Hangzhou, China, 11–13 November 2017; pp. 198–203.
- 39. Goodvin, C.; Park, E.J.; Huang, K.; Sakaki, K. Development of a real-time three-dimensional spinal motion measurement system for clinical practice. *Med. Biol. Eng. Comput.* **2006**, *44*, 1061–1075. [CrossRef]
- 40. Jones, R.; Gregory, R.; Jones, E.; Kerr, D.; Allison, S.; McLeod, A.; Titterington, D.; Hedley, A. The quality and relevance of peripheral neuropathy data on a diabetic clinical information system. *Diabet. Med.* **1992**, *9*, 934–937. [CrossRef]
- 41. Warmerdam, E.; Hausdorff, J.M.; Atrsaei, A.; Zhou, Y.; Mirelman, A.; Aminian, K.; Espay, A.J.; Hansen, C.; Evers, L.J.W.; Keller, A.; et al. Long-term unsupervised mobility assessment in movement disorders. *Lancet Neurol.* **2020**, *19*, 462–470. [CrossRef]
- Ji, G.; Msigwa, C.; Bernard, D.; Lee, G.; Woo, J.; Yun, J. Health24: Health-related Data Collection from Wearable and Mobile Devices in Everyday Lives. In Proceedings of the 2023 IEEE International Conference on Big Data and Smart Computing (BigComp), Jeju, Republic of Korea, 13–16 February 2023; pp. 336–337.
- 43. Cuesta-Vargas, A.I.; Galan-Mercant, A.; Williams, J.M. The use of inertial sensors system for human motion analysis. *Phys. Ther. Rev.* **2010**, *15*, 462–473. [CrossRef]
- 44. Tao, W.; Liu, T.; Zheng, R.; Feng, H. Gait analysis using wearable sensors. Sensors 2012, 12, 2255–2283. [CrossRef] [PubMed]
- 45. Pan, Q.; Chen, Y.; Ma, X.; Wang, C.; Chen, W. Application of Wearable Technologies in Fall Risk Assessment and Improvement in Patients with Peripheral Neuropathy: A Systematic Review. J. Sens. 2023, 2023, 1746536. [CrossRef]
- Anwary, A.R.; Yu, H.; Vassallo, M. Gait Evaluation Using Procrustes and Euclidean Distance Matrix Analysis. *IEEE J. Biomed. Health Inform.* 2019, 23, 2021–2029. [CrossRef] [PubMed]
- Wang, C.; Li, Z.; Sarpong, B. Multimodal adaptive identity-recognition algorithm fused with gait perception. *Big Data Min. Anal.* 2021, *4*, 223–232. [CrossRef]
- Talha, M.; Soomro, H.A.; Naeem, N.; Ali, E.; Kyrarini, M. Human Identification Using a Smartphone Motion Sensor and Gait Analysis. In Proceedings of the PETRA '22: The15th International Conference on PErvasive Technologies Related to Assistive Environments, Corfu, Greece, 29 June–1 July 2022; pp. 197–202.
- 49. Wang, L.; Sun, Y.; Li, Q.; Liu, T.; Yi, J. IMU-Based Gait Normalcy Index Calculation for Clinical Evaluation of Impaired Gait. *IEEE J. Biomed. Health Inform.* **2021**, *25*, 3–12. [CrossRef]
- Chen, S.; Kang, L.; Lu, Y.; Wang, N.; Lu, Y.; Lo, B.; Yang, G.-Z. Discriminative Information Added by Wearable Sensors for Early Screening—A Case Study on Diabetic Peripheral Neuropathy. In Proceedings of the 2019 IEEE 16th International Conference on Wearable and Implantable Body Sensor Networks (BSN), Chicago, IL, USA, 19–22 May 2019; pp. 1–4.
- 51. Cohen, H.S.; Mulavara, A.P.; Peters, B.T.; Sangi-Haghpeykar, H.; Kung, D.H.; Mosier, D.R.; Bloomberg, J.J. Sharpening the tandem walking test for screening peripheral neuropathy. *South Med. J.* **2013**, *106*, 565–569. [CrossRef] [PubMed]
- 52. Esser, P.; Collett, J.; Maynard, K.; Steins, D.; Hillier, A.; Buckingham, J.; Tan, G.D.; King, L.; Dawes, H. Single Sensor Gait Analysis to Detect Diabetic Peripheral Neuropathy: A Proof of Principle Study. *Diabetes Metab. J.* 2018, 42, 82–86. [CrossRef]
- 53. Wang, L.; Sun, Y.; Li, Q.; Liu, T.; Yi, J. Two Shank-Mounted IMUs-Based Gait Analysis and Classification for Neurological Disease Patients. *IEEE Robot. Autom. Lett.* 2020, *5*, 1970–1976. [CrossRef]
- 54. Abdul Razak, A.H.; Zayegh, A.; Begg, R.K.; Wahab, Y. Foot plantar pressure measurement system: A review. *Sensors* **2012**, *12*, 9884–9912. [CrossRef]

- 55. Volmer-Thole, M.; Lobmann, R. Neuropathy and Diabetic Foot Syndrome. Int. J. Mol. Sci. 2016, 17, 917. [CrossRef]
- 56. Cade, W.T. Diabetes-related microvascular and macrovascular diseases in the physical therapy setting. *Phys. Ther.* **2008**, *88*, 1322–1335. [CrossRef]
- 57. Majumder, S.; Mondal, T.; Deen, M.J. Wearable Sensors for Remote Health Monitoring. Sensors 2017, 17, 130. [CrossRef] [PubMed]
- Schlachetzki, J.C.M.; Barth, J.; Marxreiter, F.; Gossler, J.; Kohl, Z.; Reinfelder, S.; Gassner, H.; Aminian, K.; Eskofier, B.M.; Winkler, J.; et al. Wearable sensors objectively measure gait parameters in Parkinson's disease. *PLoS ONE* 2017, 12, e0183989. [CrossRef]
- Cao, Z.; Wang, F.; He, Y.; Zhang, Y.; Zhang, J. Analysis of plantar pressure in elderly diabetic patients with peripheral neuropathy. In Proceedings of the 2021 International Conference on Public Health and Data Science (ICPHDS), Chengdu, China, 9–11 July 2021; pp. 184–187.
- Corpin, R.R.A.; Guingab, H.A.R.; Manalo, A.N.P.; Sampana, M.L.B.; Abello, A.N.A.; Cruz, A.R.D.; Roxas, E.A.; Suarez, C.G.; Serrano, K.K.D. Prediction of Diabetic Peripheral Neuropathy (DPN) using Plantar Pressure Analysis and Learning Models. In Proceedings of the 2019 IEEE 11th International Conference on Humanoid, Nanotechnology, Information Technology, Communication and Control, Environment, and Management (HNICEM), Laoag, Philippines, 29 November–1 December 2019; pp. 1–6.
- Wang, D.; Ouyang, J.; Zhou, P.; Yan, J.; Shu, L.; Xu, X. A Novel Low-Cost Wireless Footwear System for Monitoring Diabetic Foot Patients. *IEEE Trans. Biomed. Circuits Syst.* 2021, 15, 43–54. [CrossRef] [PubMed]
- 62. Botros, F.S.; Taher, M.F.; ElSayed, N.M.; Fahmy, A.S. Prediction of diabetic foot ulceration using spatial and temporal dynamic plantar pressure. In Proceedings of the 2016 8th Cairo International Biomedical Engineering Conference (CIBEC), Cairo, Egypt, 15–17 December 2016; pp. 43–47.
- Voulgari, C.; Tentolouris, N.; Stefanadis, C. The ECG vertigo in diabetes and cardiac autonomic neuropathy. *Exp. Diabetes Res.* 2011, 2011, 687624. [CrossRef]
- 64. Serhiyenko, V.A.; Serhiyenko, A.A. Cardiac autonomic neuropathy: Risk factors, diagnosis and treatment. *World J. Diabetes* **2018**, *9*, 1–24. [CrossRef] [PubMed]
- Morshed, M.G.; Mukit, M.A.; Ahmed, K.I.U.; Mostafa, R.; Parveen, S.; Khandoker, A.H. Heart rate variability analysis for diagnosis of diabetic peripheral neuropathy. In Proceedings of the 2020 IEEE Region 10 Symposium (TENSYMP), Dhaka, Bangladesh, 5–7 June 2020; pp. 1253–1256.
- 66. Vinik, A.I.; Erbas, T.; Casellini, C.M. Diabetic cardiac autonomic neuropathy, inflammation and cardiovascular disease. *J. Diabetes Investig.* **2013**, *4*, 4–18. [CrossRef] [PubMed]
- 67. Bridge, P.D.; Sawilowsky, S.S. Increasing physicians' awareness of the impact of statistics on research outcomes: Comparative power of the t-test and and Wilcoxon Rank-Sum test in small samples applied research. *J. Clin. Epidemiol.* **1999**, *52*, 229–235. [CrossRef]
- Jelinek, H.F.; Cornforth, D.J.; Kelarev, A.V. Machine Learning Methods for Automated Detection of Severe Diabetic Neuropathy. J. Diabet. Complicat. Med. 2016, 1, 1000108. [CrossRef]
- 69. Herbert Jelinek, C.W.; Tinley, P. An innovative multi-disciplinary diabetes complications screening program in a rural community: A description and preliminary results of the screening. *Aust. J. Prim. Health* **2006**, *12*, 14–20. [CrossRef]
- Tung, R.T. Electrocardiographic Limb Leads Placement and Its Clinical Implication: Two Cases of Electrocardiographic Illustrations. *Kans. J. Med.* 2021, 14, 229–230. [CrossRef] [PubMed]
- Pafili, K.; Trypsianis, G.; Papazoglou, D.; Maltezos, E.; Papanas, N. Simplified Diagnosis of Cardiovascular Autonomic Neuropathy in Type 2 Diabetes Using Ewing's Battery. *Rev. Diabet Stud.* 2015, *12*, 213–219. [CrossRef]
- Hossin, M.; Sulaiman, M.N. A Review on Evaluation Metrics for Data Classification Evaluations. Int. J. Data Min. Knowl. Manag. Process 2015, 5, 1–11.
- 73. Sharanya, S.; Sridhar, P.A. A model for early diagnosis of Cardiac Autonomic Neuropathy (CAN). J. Phys. Conf. Ser. 2021, 2089, 012053. [CrossRef]
- Aminian, K.; Najafi, B.; Büla, C.; Leyvraz, P.-F.; Robert, P. Spatio-temporal parameters of gait measured by an ambulatory system using miniature gyroscopes. J. Biomech. 2002, 35, 689–699. [CrossRef] [PubMed]
- 75. Pan, J.; Tompkins, W.J. A real-time QRS detection algorithm. IEEE Trans. Biomed Eng. 1985, 32, 230–236. [CrossRef]
- 76. Luo, H.; Gao, B. Development of smart wearable sensors for life healthcare. Eng. Regen. 2021, 2, 163–170. [CrossRef]
- 77. Heikenfeld, J.; Jajack, A.; Rogers, J.; Gutruf, P.; Tian, L.; Pan, T.; Li, R.; Khine, M.; Kim, J.; Wang, J.; et al. Wearable sensors: Modalities, challenges, and prospects. *Lab Chip* **2018**, *18*, 217–248. [CrossRef]
- 78. Sejdic, E.; Lowry, K.A.; Bellanca, J.; Redfern, M.S.; Brach, J.S. A comprehensive assessment of gait accelerometry signals in time, frequency and time-frequency domains. *IEEE Trans. Neural Syst. Rehabil. Eng.* **2014**, 22, 603–612. [CrossRef]
- 79. Fusco, N.; Cretual, A. Instantaneous treadmill speed determination using subject's kinematic data. *Gait Posture* **2008**, *28*, 663–667. [CrossRef]
- Khandakar, A.; Mahmud, S.; Chowdhury, M.E.H.; Reaz, M.B.I.; Kiranyaz, S.; Bin Mahbub, Z.; Ali, S.H.M.; Bakar, A.A.A.; Ayari, M.A.; Alhatou, M.; et al. Design and Implementation of a Smart Insole System to Measure Plantar Pressure and Temperature. *Sensors* 2022, 22, 7599. [CrossRef]
- Kukreja, G.S.; Alok, A.; Reddy, A.K.; Nersisson, R. IoT Based Foot Neuropathy Analysis and Remote Monitoring of Foot Pressure and Temperature. In Proceedings of the 2020 5th International Conference on Computing, Communication and Security (ICCCS), Patna, India, 14–16 October 2020; pp. 1–6.

- Sempere-Bigorra, M.; Brognara, L.; Julian-Rochina, I.; Mazzotti, A.; Cauli, O. Relationship between deep and superficial sensitivity assessments and gait analysis in diabetic foot patients. *Int. Wound J.* 2023, 20, 3023–3034. [CrossRef] [PubMed]
- 83. Barrot, M. Tests and models of nociception and pain in rodents. *Neuroscience* 2012, 211, 39–50. [CrossRef] [PubMed]
- 84. Al-Geffari, M. Comparison of different screening tests for diagnosis of diabetic peripheral neuropathy in Primary Health Care setting. *Int. J. Health Sci.* 2012, *6*, 127–134. [CrossRef]
- 85. Massey, F.J. The Kolmogorov-Smirnov Test for Goodness of Fit. J. Am. Stat. Assoc. 1951, 46, 68–78. [CrossRef]
- Veličković, Z.; Dolijanovic, S.P.; Tomonjic, N.; Janjić, S.; Stojic, B.; Radunovic, G. AB1524 a novel accelerometry-based method for early detection of peripheral neuropathy associated with systemic autoimmune rheumatic diseases. *BMJ* 2023, *82* (Suppl. 1), 1994–1995.
- Sabry, F.; Eltaras, T.; Labda, W.; Alzoubi, K.; Malluhi, Q. Machine Learning for Healthcare Wearable Devices: The Big Picture. J. Healthc. Eng. 2022, 2022, 4653923. [CrossRef]
- Rodriguez-León, C.; Villalonga, C.; Munoz-Torres, M.; Ruiz, J.R.; Banos, O. Mobile and Wearable Technology for the Monitoring of Diabetes-Related Parameters: Systematic Review. *JMIR Mhealth Uhealth* 2021, 9, e25138. [CrossRef]
- 89. Banerjee, S.; Hemphill, T.; Longstreet, P. Wearable devices and healthcare: Data sharing and privacy. *Inf. Soc.* **2017**, *34*, 49–57. [CrossRef]
- Sui, A.; Sui, W.; Liu, S.; Rhodes, R. Ethical considerations for the use of consumer wearables in health research. *Digit. Health* 2023, 9, 20552076231153740. [CrossRef]
- Ganapathi, S.; Palmer, J.; Alderman, J.E.; Calvert, M.; Espinoza, C.; Gath, J.; Ghassemi, M.; Heller, K.; Mckay, F.; Karthikesalingam, A.; et al. Tackling bias in AI health datasets through the STANDING Together initiative. *Nat. Med.* 2022, 28, 2232–2233. [CrossRef]
- Bharati, S.; Mondal, M.R.H.; Podder, P. A Review on Explainable Artificial Intelligence for Healthcare: Why, How, and When? IEEE Trans. Artif. Intell. 2023, 1–15. [CrossRef]
- Vijayan, V.; Connolly, J.P.; Condell, J.; McKelvey, N.; Gardiner, P. Review of Wearable Devices and Data Collection Considerations for Connected Health. Sensors 2021, 21, 5589. [CrossRef] [PubMed]
- 94. Tropschuh, B.; Windecker, S.; Reinhart, G. Study-based evaluation of accuracy and usability of wearable devices in manual assembly. *Prod. Manuf. Res.* 2022, *10*, 569–582. [CrossRef]
- 95. Boateng, G.; Motti, V.G.; Mishra, V.; Batsis, J.A.; Hester, J.; Kotz, D. Experience: Design, Development and Evaluation of a Wearable Device for mHealth Applications. *Proc. Annu. Int. Conf. Mob. Comput. Netw.* **2019**, 2019, 31.
- Ha, S.; Park, S.; Lim, H.; Baek, S.H.; Kim, D.K.; Yoon, S.-H. The placement position optimization of a biosensor array for wearable healthcare systems. J. Mech. Sci. Technol. 2019, 33, 3237–3244. [CrossRef]
- 97. Mankins, J.C. Technology Readiness Levels; NASA: Washington, DC, USA, 6 April 1995.
- Krajcsik, J.R. The State of Health Data Privacy, and the Growth of Wearables and Wellness Apps. Master's Thesis, University of Pittsburgh, Pittsburgh, PA, USA, 2022.
- 99. Canali, S.; Schiaffonati, V.; Aliverti, A. Challenges and recommendations for wearable devices in digital health: Data quality, interoperability, health equity, fairness. *PLOS Digit. Health* **2022**, *1*, e0000104. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.